Preoperative intravenous iron for anaemia in elective major open abdominal surgery: the PREVENTT RCT

Toby Richards,^{1,2,3*} Ravishankar Rao Baikady,⁴ Ben Clevenger,^{3,11} Anna Butcher,³ Sandy Abeysiri,^{2,3} Marisa Chau,^{2,3} Rebecca Swinson,⁵ Tim Collier,⁵ Matthew Dodd,⁵ Laura Van Dyck,⁵ Iain Macdougall,⁶ Gavin Murphy,⁷ John Browne,⁸ Andrew Bradbury⁹ and Andrew Klein¹⁰

¹Division of Surgery, University of Western Australia, Perth, WA, Australia ²Institute of Clinical Trial and Methodology, University College London, London, UK ³Division of Surgery, University College London, London, UK

- ⁴The Royal Marsden NHS Foundation Trust, London, UK
- ⁵Clinical Trials Unit, London School of Hygiene & Tropical Medicine, London, UK ⁶Department of Renal Medicine, King's College Hospital, London, UK
- ⁷NIHR Leicester Biomedical Research Centre, Department of Cardiovascular Sciences, University of Leicester, Leicester, UK
- ⁸School of Public Health, University College Cork, Cork, Ireland
- ⁹University Department of Vascular Surgery (University of Birmingham), Solihull Hospital, Solihull, UK
- ¹⁰Department of Anaesthesia and Intensive Care, Royal Papworth Hospital, Cambridge, UK
- ¹¹Department of Anaesthesia, Royal National Orthopaedic Hospital, Stanmore, UK

*Corresponding author toby.richards@uwa.edu.au

Declared competing interests of authors: Andrew Klein reports grants and personal fees from Pharmacosmos A/S (Holbaek, Denmark) and Fisher and Paykel (Auckland, New Zealand), personal fees from Vifor Pharma (Glattbrugg, Switzerland) and Masimo (Irvine, CA, USA), and grants from Haemonetics® (Braintree, MA, USA) and HemoSonics (Charlottesville, VA, USA) outside the submitted work. Iain Macdougall reports grants and personal fees from Vifor Pharma outside the submitted work. Ravishankar Rao Baikady reports grants and other support from Pharmacosmos UK (Reading, UK), Vifor Pharma, Novo Nordisk A/S (Bagsværd, Denmark) and Fresenius Kabi (Bad Homburg, Germany) outside the submitted work. Gavin Murphy reports grants from Zimmer Biomet (Warsaw, IN, USA) and other from Terumo Corporation (Tokyo, Japan) outside the submitted work, and is part of Clinical Trials Units (CTUs) funded by the National Institute for Health Research (NIHR) CTU Standing Advisory Committee. Toby Richards reports grants from NIHR Health Technology Assessment (HTA) during the conduct of the study, grants, personal fees and non-financial support from Pharmocosmos and Vifor Pharma, and personal fees from Tiash Ltd (Republic of Singapore) outside the submitted work. Sandy Abeysiri reports grants from NIHR HTA during the conduct of the study and has been working on PREVENTT as part of her PhD at University College London. She has also worked at The Iron Clinic (London, UK), which provides intravenous iron therapy. Andrew Bradbury reports membership of the NIHR HTA Interventional Procedures Panel, NIHR HTA Prioritisation Group, NIHR HTA Remit and Competitiveness Committee and NIHR HTA Hospital Based Care Methods Group (2014–2020).

Published February 2021 DOI: 10.3310/hta25110

Plain English summary

The PREVENTT RCT

Health Technology Assessment 2021; Vol. 25: No. 11 DOI: 10.3310/hta25110

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

n patients undergoing major surgery, anaemia (low blood count) is a common problem. Anaemia is often a consequence of the disease necessitating surgery and can make people feel tired and unwell. Anaemia increases the need for a blood transfusion at the time of surgery, and patients with anaemia have more complications from surgery, prolonged hospital stay and delayed recovery.

Iron deficiency is the most common cause of anaemia. An iron infusion has been shown to be effective to rapidly treat anaemia, but it is not known if this is effective in treating anaemia in the presurgical setting, and whether or not this may benefit patients.

The main aim of this study was to assess if intravenous iron can treat anaemia in patients before major surgery and if this will reduce the need for blood transfusion, make patients feel better and, consequently, help them do better during and after surgery.

A total of 487 patients with anaemia were recruited from 46 UK hospitals before major abdominal surgery. Patients were randomly allocated to receive an infusion of iron or placebo 10 days to 6 weeks before their surgery. Patients were followed up at 8 weeks and 6 months after their surgery.

Intravenous iron increased the blood count in patients before surgery. There was no difference in blood transfusion rates or patient deaths between those who received intravenous iron and those who received placebo. Similarly, there was no difference in the patients' postoperative complications or length of hospital stay. Patients who received iron had a higher blood count at 8 weeks and 6 months post operation and there were fewer re-admissions to hospital for complications.

In conclusion, for patients undergoing major surgery, giving intravenous iron to treat anaemia before the operation did not reduce the need for blood transfusion. Further work is needed to evaluate whether or not there is any benefit after discharge from hospital.

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 3.370

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, the Cochrane Library and Clarivate Analytics Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 10/104/06. The contractual start date was in September 2012. The draft report began editorial review in November 2019 and was accepted for publication in July 2020. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care.

Copyright © 2021 Richards *et al.* This work was produced by Richards *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk